

Integrated Power and Thermal Management Systems for Civil Aircraft: Review, Challenges, and Future Opportunities

Zeyu Ouyang *🗅, Theoklis Nikolaidis 🗅 and Soheil Jafari 🕒

Centre for Propulsion and Thermal Power Engineering, School of Aerospace Transport and Manufacturing (SATM), Cranfield University, Cranfield MK43 0AL, UK; t.nikolaidis@cranfield.ac.uk (T.N.) * Correspondence: z.ouyang@cranfield.ac.uk

Abstract: Projects related to green aviation designed to achieve fuel savings and emission reductions are increasingly being established in response to growing concerns over climate change. Within the aviation industry, there is a growing trend towards the electrification of aircraft, with more-electric aircraft (MEA) and all-electric aircraft (AEA) being proposed. However, increasing electrification causes challenges with conventional thermal management system (TMS) and power management system (PMS) designs in aircraft. As a result, the integrated power and thermal management system (IPTMS) has been developed for energy-optimised aircraft projects. This review paper aims to review recent IPTMS progress and explore potential design solutions for civil aircraft. Firstly, the paper reviews the IPTMS in electrified propulsion aircraft (EPA), presenting the architectures and challenges of the propulsion systems, the TMS cooling strategies, and the power management optimisation. Then, several research topics in IPTMS are reviewed in detail: architecture design, power management optimisation, modelling, and analysis method development. Through the review of state-of-the-art IPTMS research, the challenges and future opportunities and requirements of IPTMS design are discussed. Based on the discussions, two potential solutions for IPTMS to address the challenges of civil EPA are proposed, including the combination of architecture design and power management optimisation and the combination of modelling and analysis methods.

Keywords: integrated thermal and power management systems; thermal management systems; power management systems; more-electric aircraft; energy-optimised aircraft

1. Introduction

Aircraft engines generate greenhouse gases, such as carbon dioxide (CO₂) and nitrogen oxides (NO_x) [1]. The aviation industry was responsible for about 3% of global CO₂ emissions in 2019, and the CO₂ emissions could increase 2.5 times from 2015 to 2050 without rapid technological advancements and stronger policies [2]. Global and regional organisations have focused on these challenges. The United Nations has stated goals to combat climate change [3]. In the USA, NASA proposed a strategic implementation plan to reduce NO_x emissions by 80%, fuel consumption by 80%, and noise by 42 dB by 2035 [4]. In Europe, the Advisory Council for Aeronautics Research in Europe (ACARE) has set targets under the 'Fly the Green Deal' initiative to reduce net CO₂ emissions by 55% compared to the 1990 baseline and to achieve net-zero CO₂ emissions by 2050 [5]. If an aircraft is more electric, it requires less carbon-based fuel, potentially resulting in less CO₂ and NO_x emissions, high efficiency and reliability, and less noise [6]. As a result, there is a growing trend toward the electrification of aircraft with the proposed more-electric aircraft (MEA) and all-electric aircraft (AEA).

In conventional aircraft, an engine provides four main types of power: (1) mechanical, (2) hydraulic, (3) pneumatic, and (4) electric. These types of power are used for critical components in conventional aircraft systems, including pumps (hydraulic/mechanical), actuation systems (pneumatic), environmental control systems (ECSs) (pneumatic), wing



Citation: Ouyang, Z.; Nikolaidis, T.; Jafari, S. Integrated Power and Thermal Management Systems for Civil Aircraft: Review, Challenges, and Future Opportunities. *Appl. Sci.* 2024, *14*, 3689. https://doi.org/ 10.3390/app14093689

Academic Editor: Rosario Pecora

Received: 29 March 2024 Revised: 20 April 2024 Accepted: 24 April 2024 Published: 26 April 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). ice protection systems (WIPS) (pneumatic), and avionics (electric). The aim of MEA/AEA is to reduce and ultimately eliminate all but electric power in aircraft. The B787, a civil MEA, for the first time eliminates the use of bleed air, implementing electric power for WIPS instead of pneumatic [7,8]. However, its propulsive power is still not electric. In an AEA, not only is the power of aircraft systems electric, but the propulsive power is also provided by electrochemical energy units such as batteries and fuel cells. Compared to MEA, AEA have lower operating costs and no emissions because all forms of power are electric.

MEA/AEA have obvious advantages, but the increasing use of electric power also brings more challenges. For military aircraft, the use of electrified subsystems such as ECSs, avionics, weapons, and actuators, as well as associated heat losses, has risen substantially [9,10], dramatically increasing the demand for heat sinks [11]. However, the limited heat sinks in military aircraft result in technical obstacles for an aircraft's thermal management system (TMS). In civil aircraft, not only is more and more of the power of subsystems electric, but the propulsion of aircraft is also increasingly electrified, with a trend toward hybrid-electric aircraft (HEA) and AEA. This growing use of electric power in civil aircraft also causes thermal challenges for airframe materials and heat sink capacity. The highly electrified subsystems increase thermal requirements and result in challenges for the power management system (PMS). In an aircraft, the PMS needs to provide enough power for both cooling and critical electric components. Particularly for electrified propulsion aircraft (EPA), such as HEA and AEA, it is necessary to split the power to provide both cooling power and propulsive power. These complicated interactions between the TMS and PMS therefore imply that the development of energy management in only one energy domain (thermal or power) would make it hard to optimise energy consumptions for future MEA/AEA.

How to manage electric and thermal energies has become a critical question for researchers to minimise the challenges offsetting the benefits offered by EPA. As a result, a novel aircraft energy management approach was proposed, which is the integrated power and thermal management system (IPTMS). Many institutions, such as Honeywell, the European Space Agency, and the United States Air Force, have focused on energy-optimised aircraft projects and developed IPTMSs for advanced thermal energy design. The United Kingdom's Aerospace Technology Institute (ATI) has listed novel thermal management systems for EPA as their priority study strategy [12]. It is highly likely that an IPTMS is a big step toward MEA/AEA and will play a more important role in energy management for aircraft [13].

There are published papers that have reviewed relevant fields of TMSs or PMSs, but there are limited sources in the public domain on the integrated system of a PMS and TMS. The authors in [14] reviewed the growing challenges of TMSs for military and civil aircraft due to increasing electrification and discussed future opportunities and potential solutions based on a simple classification of TMS factors and research topics. This article primarily focuses on the TMS for conventional aircraft and did not explore more details of an IPTMS for civil electrical propulsion aircraft. In [15], the authors review the TMS developments for HEA. They focused on different TMS architectures for the key components in electrical propulsion system to investigate an optimised TMS architecture under different operating conditions. Furthermore, a comprehensive review of the TMS challenges for hybridelectric aircraft is presented in [16]. They investigated the challenges and solutions for the TMS components and subsystems to identify the future opportunities at the component level. The authors in [17] reviewed the architectures, control system design, and energy management strategies for electric propulsion systems to reduce fuel burn and emissions and presented an overview of future trends in aviation electrification. While the authors presented a wide literature review of energy management strategies, they did not review the strategies for managing energy flows in thermal and power domains.

Previous research has explored thermal and power management for Electrical Propulsion Aircraft (EPA), yet a gap exists in the literature regarding various types of IPTMS research for potential solutions in civil EPA. This paper aims to address this gap by reviewing recent IPTMS progress and exploring potential design solutions for civil aircraft. The paper is organised as follows. Section 2 presents the IPTMS in EPA, introducing the architectures and challenges of EPA, TMS cooling strategies, and power management optimisation approaches. Section 3 focuses on the architecture design of an IPTMS in two domains: (1) an IPTMS with energy recovery and (2) an IPTMS with an advanced electrical propulsion system (EPS) and TMS. Section 4 describes the power management (PM) design of the IPTMS. Sections 5 and 6 present the modelling and analysis methods of the IPTMS. Section 7 discusses the challenges and future opportunities of IPTMS, suggesting the requirements for future IPTMS design. Finally, the conclusion is presented in Section 8.

2. Integrated Power and Thermal Management in Electrified Propulsion Aircraft

2.1. Integrated Power and Thermal Management System

Integrated Power and Thermal Management (IPTMS) is a system/approach to manage the energy flows in two domains: (1) electric and (2) thermal. In simple terms, it can be assumed as an integration of a TMS and PMS. When it plays the role of TMS, it rejects heat from heat sources to ensure that components operate at their operating temperatures to avoid degradation. On the other hand, when it contributes to power management (PM), it allocates and balances the power to minimise it while providing sufficient power to critical power loads for safety [18].

The IPTMS has been widely researched and has different labels in different papers, such as power and thermal management system (PTMS), electro-thermal management system, energy-thermal management system, etc. In this paper, IPTMS is selected as it implies an integrated system to manage electric and thermal energies.

2.2. Electrified Propulsion Aircraft

The definition of degree of hybridisation (DoH) developed by Isikveren et al. [19] helps to categorise the types of aircraft (shown in Table 1) as follows:

• DoH for power (H_P)

$$\bigcirc$$
 $H_P = \frac{P_{motor}}{P_{tot}}$

$$\bigcirc P_{tot} = P_{motor} + P_{engine}$$

• DoH for energy (H_E)

$$\begin{array}{ll} \bigcirc & H_E = \frac{E_{bat}}{E_{tot}} \\ \bigcirc & E_{tot} = E_{battery} + E_{engine} \end{array}$$

where P_{motor} = power of motor, P_{engine} = power of engine, P_{tot} = aircraft total power, $E_{battery}$ = energy of battery, E_{engine} = energy of engine, and E_{tot} = aircraft total energy.

Table 1. Classification of aircraft.

Aircraft Type	DoH for Power	DoH for Energy
Conventional aircraft	0	0
Series HEA	1	<1
Turboelectric	>0	0
Parallel HEA	<1	<1
AEA	1	1

Based on the DoH, the aircraft propulsion system can be electrified using motors $(H_P > 1)$ and without a battery (e.g., turboelectric aircraft). Figure 1 shows the simplified schematic of the propulsion system for EPA. Readers can refer to [20–23] for more information on electrified aircraft propulsion.



Figure 1. Simplified schematic of the propulsion system of EPA (adapted from [22]) (DC: direct current, GBx: power gearbox, and GT: gas turbine).

Generally, as Figure 1 shows, an electrical propulsion system (EPS) consists of electric machines (motors or generators), power distribution systems (DC or AC bus), and converters (rectifiers, inverters, or DC/DC converter). The motors have four main types of loss [24]: (1) ohmic losses, (2) core losses, (3) windage losses, and (4) other losses. The other losses are usually assumed to be constant as they are independent on the motor torque or rotational speed. Assuming the loss coefficients remain the same, the ohmic losses are dependent on the motor's torque, while the core losses and the windage losses are dependent on the motor's rotational speed. In this case, when the rotational speeds of the aircraft's propulsors are constant, the heat from the motors might not be reduced significantly even under low-power conditions, which increases the requirements of the TMS. Additionally, with the advancement of motor technology, motors are becoming smaller and lighter. However, a smaller size can limit the size of the motor's cold plate. Even though the efficiency of the motor may improve, under high-power conditions, the heat flux of the motor could increase, introducing challenges for the TMS. For the power distribution systems, the higher the system voltage, the less heat from the bus is generated. However, the higher system voltage might require a bigger size of battery or generator, which requires trade-off studies between the EPS and the TMS designs. Compared with other electrical components, the range of the operating temperature of the battery is narrower, usually within 15–45 °C [25]. While ensuring the battery temperature does not exceed the maximum temperature, an active temperature control might be required under low-temperature conditions to ensure the temperature is above the minimum temperature. Furthermore, the battery's state of charge should be above the minimum state of charge to prevent damage. Therefore, the IPTMS design should consider the power and thermal requirements of the battery.

For series and parallel HEA, the battery power output can be determined by determining the DoH, or by determining the power of the gas turbine engines. For example, the DoH for each flight phase is determined, and then the battery is sized. Alternatively, the maximum power output of the gas turbine engines is determined so that during the take-off flight segment, the supplementary power is provided by the battery. Additionally, at the end of the mission (i.e., during the descent or landing flight phase), the battery could also provide power, or it could be charged by the engine-driven generators to save energy. For AEA, all the power, such as for cooling, propulsion, emergency, etc., are provided by the battery. The battery is sized to ensure that the power requirements are satisfied, and it is necessary to ensure that the battery's state of charge is above its minimum state of charge to prevent damage. For turboelectric aircraft, the engines drive the generators to

5 of 19

provide power. The power is distributed to several motors, which then provide propulsion. Similar to AEA, all the electric power comes from the turboelectric aircraft's turbogenerator which is the assembly of the engine and the generator. Additionally, the turbogenerator can also provide propulsive power through the gas turbine engine, so trade-off studies or optimisation strategies are required to balance the electric power and the propulsive power from the engines to minimise fuel consumption. Overall, to guarantee the green and sustainable nature of EPA, trade-off studies, optimisation strategies, implementation of high-efficiency electrical components, and so on are required. Additionally, battery power supply strategies can optimise energy use and maintain batteries, thereby extending battery life. The handling of expired batteries, such as through recycling or reuse, is related to battery technologies or relevant component-level studies, which are outside the scope of this paper.

2.3. Thermal Management System

In TMS, three key elements should be considered: heat acquisition, heat transport, and heat rejection [26]. TMS absorbs the heat from heat sources (heat acquisition) and then transfers the heat to coolants. Finally, through the coolants of cooling loops (heat transport), the heat is rejected to terminal heat sinks (heat rejection).

As discussed in Section 2.2, because of additive electric components, the considerations of TMS for conventional aircraft and electrical propulsion aircraft (EPA) are different. The TMS design of a conventional aircraft typically does not consider turbine blade cooling and relevant mechanisms [27]. Therefore, the heat loads considered by the TMS for a conventional aircraft include accessory gearboxes, engine gearboxes, engine-driven generators, bearings of shafts, oil pumps, etc., [27–29]. Compared with conventional aircraft, the additive heat loads of EPA involve converters (including inverters and rectifiers), batteries, electric bus/feeders, motors, and generators. These additive heat loads cause challenges for system weight, efficiency, power management, and thermal management.

Generally, there are three active TMS cooling strategies to transfer heat: (1) air cooling, (2) liquid cooling, and (3) vapour cycle system (VCS)/phase-change cooling. The most common method of cooling is air cooling, which is mature and widely used in civil aircraft [30]. Many coolants for liquid cooling have been studied, including the coolants that have been applied in aircraft such as fuel, oil, water/glycol mixture, polyalphaolephin (PAO), etc. [31], as well as coolants for potential application in aircraft such as supercritical carbon dioxide (sCO₂), therminol fluids, etc. [32–35]. Compared with air cooling and liquid cooling, the coolants of VCS cooling are two-phase. Because of the phase-change coolants, VCS cooling has better efficiency but is heavier and less commercially matured [36,37]. Active air cooling is usually the lightest, because it is open-loop and does not require heat exchangers, coolant pumps, and tanks. However, under high-heat-load conditions, the ram air drag, the weight, and the power due to the air-cooling fan could be considerable. In this case, air cooling is no longer suitable. In terms of liquid cooling, the IPTMS design should consider the coolant performance in thermal and power domains. When a water/glycol mixture is selected as the coolant, the use of propylene glycol can increase the freezing point of the coolant but can increase the viscosity of the coolant, which causes more cooling power for overcoming the pressure drop. Additionally, the IPTMS design should consider how different coolants impact the system weight, power requirements, and induced drag under high- and low-temperature conditions. As three active TMS cooling strategies have their pros and cons, the IPTMS design should compare their performance at the system level and obtain the optimal strategy.

The heat sinks for aircraft typically involve fuel and bleed/ram air. Using fuel as the terminal heat sink for an aircraft can improve its engine efficiency as the fuel is preheated. Additionally, unlike other coolants, the coolant of fuel does not require additive components (e.g., coolant pumps and tanks). However, one should be careful that the fuel is not overheated and that the flow rate of fuel is sufficient for heat transfer. This is why HEA

usually use ram air as the terminal heat sink instead of fuel. The heat sink of air is low-cost, simple, and effective [33,35], but it induces drag [36,38].

Passive cooling can be implemented in a TMS. Compared with active cooling, passive cooling does not require any power for cooling. Generally, the passive cooling could include natural air convection, radiation, a heat acquisition technique [39,40], and aircraft skin heat exchangers [41–44]. Although natural air convection is less suitable for hightemperature heat loads [30,45], under low-temperature conditions, such as high-altitude, low-power, or cold-day conditions, the air's natural convection and radiation are potentially sufficient for cooling. The natural air convection and radiation are related to the volume of the component and the spacing between the heat sources. The bigger the component's volume, the larger the heat transfer area for natural air convection and radiation. At the same time, the bigger volume causes heavier weight and challenges in installation. In this case, it is necessary to assess the components' performance accurately to investigate the performance of natural air convection and radiation under the aircraft's operation conditions. Additionally, the design parameters, such as the design power of the motor and the design capacity of the battery, would also affect the passive cooling performance. These design parameters depend on the power requirements for propulsion and cooling, so the IPTMS design requires trade-off studies between component sizing and passive TMS performance. For HEA, the motors and generators are usually installed next to the gas turbine engines so that the natural air convection and radiation cannot be implemented for those components due to their limited spacing. Additionally, the battery might need active temperature control to ensure its temperature is above the minimum temperature to prevent damage. Therefore, the benefits of the natural air convection and radiation might have less positive impact. One of the challenges for skin heat exchanger applications is the heat transport from the heat sources to the heat sinks. If the coolant absorbs the heat and directly transfers the heat to the aircraft skin, this passive cooling method might consume more fuel than implementing a traditional active TMS [44]. The reason for this is that the pumps still actively operate to distribute the fluid, which requires more power, causing heavier pumps. This challenge leads to a need for an aircraft outer mould line cooling technique. In this case, the heat from the component is transferred to the outer mould line. The outer mould line distributes the heat to the skin heat exchanger, and the heat is finally rejected to the ambient air. To sufficiently cool the components, the outer mould line cooling could induce weight penalty.

2.4. Power Management Optimisation

Figure 2 shows a control architecture for EPA. The control for EPA is generally recognised as two levels: supervisory control and component control [46]. The EPA has a supervisory controller to manage the commands for the conventional and the electrified propulsion systems. Usually, in some papers, "control system design" or "controller design" refers to the component (lower)-level design, and the supervisor control approach refers to "power (energy) management strategy". In the IPTMS, based on the thermal requirements, the TMS components, such as the coolant pumps and the cooling fans, require power for cooling, which can be provided by the engine-driven generators or the battery. In the electrical propulsion system, the battery needs to provide power for the motors to drive propulsors. In this case, the supervisory controller should balance the energy flows, satisfying the power and thermal requirements, and obtain an optimal solution to minimise the energy consumption. Therefore, the power management optimisation strategies are researched for the supervisory controller.



Figure 2. Control architecture for EPA, where the green blocks represent the engine system, blue blocks represent the power management system, and the orange blocks represent the thermal management system (adapted from [46]).

PM strategies have been studied widely for hybrid-electric ground-based vehicles (HEVs) and have been integrated with TMSs to manage energy flows at the vehicle level [47,48]. However, the general technologies developed for HEVs are not available for implementation in aircraft as aircraft have different considerations. Aircraft manage energy flows based on mission profiles. An aircraft mission has several segments in the following sequence: taxi-out, takeoff, climb, cruise, descent, landing, and taxi-in, and the propulsion requirements per mission segment (e.g., balancing thrust and drag) are different. In HEA, the PM strategy is more complicated, minimising fuel consumption with electric power, providing sufficient power for TMS and other electric components, and ensuring the state of charge (SOC) of the battery is above the minimum value to avoid damage [49].

PM strategies can be generally categorised into two types: causal and non-causal. Researchers apply non-causal ones when future conditions are known [50–53] and apply causal PM strategies for unknown-future conditions [54–58]. Both methods can be implemented for real-time applications, depending on if the mission profile is known beforehand or not. For aircraft, the decision of the type of PM strategy depends on the objective, optimising for the entire flight mission or a single moment. If the optimisation is non-causal, the overall fuel consumption is assessed and minimised. A non-causal cost function can be described by Equation (1). If the optimisation is causal, the fuel consumption for each time step is assessed and minimised. A causal cost function can be described by Equation (2).

$$\mathbf{J} = \int_{t_i}^{t_{end}} \mathbf{W} t_{fuel}(t_k) dt, \ t_k \in [t_i, \ t_{end}]$$
(1)

$$\mathbf{J} = W_{fuel}(t_k), \ t_k \in [t_i, \ t_{end}]$$
⁽²⁾

where J = objective function, t = time, $t_i = initial time$, $t_{end} = end time$, $Wt_{fuel} = fuel weight$, and $W_{fuel} = fuel flow rate$.

3. IPTMS Architecture Design

Generally, the IPTMS architecture design can be categorised into two approaches: (1) integrate different aircraft subsystems based on energy recovery theory, and (2) design advanced EPS and TMS architectures considering the interactions between the TMS and PMS.

3.1. IPTMS with Energy Recovery

With energy recovery, an IPTMS converts thermal energy to other secondary powers, such as mechanical power and electrical power. A simple IPTMS with energy recovery can be assumed as an integration of an APU (auxiliary power unit) and ECS (environmental control system), consisting of a compressor, a turbine, and a generator, as Figure 3 shows. In this kind of IPTMS, fluids (i.e., gas and coolant) expand within the turbine to achieve a low temperature for cooling. Through the expansion, the turbine output power can drive the compressor and the generator. Thereby, the IPTMS can have heat-cooling and power-generation functions.



Figure 3. Simple IPTMS with energy recovery.

The IPTMS achieves ECS functions by implementing thermodynamic cycles, such as the Brayton cycle and Rankine cycle. The ECS with reversed Brayton cycle principle is used widely and has become commercially mature [37]. Generally, the working fluid of this type of ECS is air, resulting in low weight, high operating reliability, and low operating cost [59,60]. Compared with the ECS based on the Brayton cycle, the ECS based on reversed Rankine cycle can eliminate the use of bleed air. Additionally, it can have better efficiency, but it is heavier and less commercially mature [37].

A prominent example of IPTMS based on the Brayton cycle is the IPTMS developed by Honeywell [13,61]. This IPTMS integrates four conventional aircraft system functions: the auxiliary power unit (APU), emergency power unit (EPU), environmental control system (ECS), and thermal management system (TMS). The system has four operating modes: the Self-Start Mode, the Main-Engine Start, the Cooling Mode, and the Emergency Power Mode. The functions of these modes are as follows:

Self-Start Mode: In this mode, the battery initially provides power for the starter/ generator of the IPTMS to drive the turbomachinery. When the shaft speed of the turbomachinery increases to an initial value, the IPTMS achieves self-sustaining operation and transitions to the Main-Engine Start Mode.

Main-Engine Start: In this mode, the system plays the role of APU. The IPTMS can provide power to start the main engine without the use of the aircraft battery.

Cooling Mode: In this mode, the IPTMS is powered by the bleed air from the main engine instead of the combustor of the IPMTS. The IPTMS plays the roles of roles of ECS and TMS, implementing active air cooling with a reversed Brayton cycle. The turbomachinery of the IPTMS functions similarly to a conventional air cycle machine, with the added capability of driving a generator to provide power, as Figure 3 shows. The IPTMS generates cooled air which not only serves the purpose of air conditioning but also aids in dissipating heat from avionics, flight-critical electronics, and oil.

Emergency Power Mode: When the main engine fails, the system transitions to the Emergency Power Mode. In this mode, the IPTMS plays the role of EPU, and is powered by the combustor of the IPTMS instead of the bleed air. The IPTMS-driven generator provides power for critical inflight electronics or for the main engine restart.

Based on this milestone set by the IPTMS of Honeywell, several studies have attempted to improve the IPTMS performance by reducing overall system weight, increasing energy conversion efficiency, and extracting more power. Dooley et al. [62,63] proposed a unit,

named the Integrated Power and Cooling Unit, for the IPTMS. This unit simplifies the system by reducing the number of recirculation compressor components and limiting the use of bleed and ram air. Although this IPTMS design decreases the system's weight and volume and minimises the negative impact of bleed and ram air, it was found that the fluid efficiency of phase-change cooling and the recovery effectiveness are low. Abolmoali et al. [64] integrated a closed-loop air cycle system with an open-loop gas generator cycle to cool the gas turbine engine heat loads and provide sufficient power for auxiliary power loads. This IPTMS can provide up to 78.5% more power than the conventional power generation approach under sea-level static conditions. However, due to the impact of the flight altitude, the IPTMS cannot provide sufficient power for auxiliary power loads under other operation conditions and still requires supplementary power from the engine-driven generators. In [65], an IPTMS with a multi-mode Rankine cycle system was developed which was able to remove 10.7 kW from the coolant and satisfy the power requirements of the auxiliary systems (230 W at steady state). However, the response time of the heat transfer is slower than the power generation. Stoia et al. [66] designed an IPTMS for highspeed aircraft based on the Brayton cycle, using sCO_2 (supercritical carbon dioxide) as the coolant. This IPTMS can be recuperated (with the use of a recuperator) or non-recuperated. The recuperation depends on the ratio of heat transport to power required of the IPTMS. The IPTMS in [66] is recuperated and can provide power density greater than 3 kWe/L with up to 35% energy conversion efficiency. However, because of the recuperator, the weight and the volume of the IPTMS are significantly increased.

The IPTMS design with energy recovery is usually applied to military aircraft because it can save aircraft system weight and volume by integrating different subsystems. Compared with the conventional air cycle machine, although this type of IPTMS still needs additional power input to drive the compressor, its turbine can drive a generator to provide power, converting thermal energy to electric energy. Therefore, the power density and the energy conversion efficiency could be the key target parameters for the IPTMS design to obtain a smaller and lighter system which provides more electric power.

3.2. IPTMS with Advanced EPS and TMS

As discussed in Section 2.2, the heat loads of EPA are different from conventional aircraft, and because of the additive electric components, the TMS design for EPA faces more challenges, such as increased weight, induced drag, and TMS power use [39,67]. Therefore, the IPTMS architecture design for EPA should consider the interactions between the EPS and TMS architectures.

Kim et al. [68] proposed a sensitivity study of IPTMS for a Megawatt-class (MWclass) turboelectric aircraft. They investigated the impacts of advancements in power distribution systems and TMSs on the aircraft's performance and proposed three potential TMS architectures. They found that a 150-passenger MW-class turboelectric aircraft is infeasible at the current technological level due to the TMS challenges. In addition, they state that the advancements in TMSs are more urgently needed for the aircraft. Chapman et al. [69,70] proposed IPTMS architectures for different aircraft implementing advanced AC power systems. They developed a high-efficiency megawatt motor and a high-efficiency AC converter for the AC transmission of the aircraft EPS. Compared with the conventional DC transmission approach, the EPS with AC transmission reduces the conversion steps and the electric component weights. As a result, the TMS is redesigned to reduce the cooling loops. It was found that their IPTMS architecture design can significantly reduce overall system weight, drag, and power use. In [71], the IPTMS of a long-endurance unmanned HEA integrates the TMS with the propulsion system through a fuel loop. The fuel is reheated before burning in the combustor, rejecting the heat from the powertrain in order to increase the engine's thermal efficiency. Compared with the conventional propulsion system, the IPTMS can save 750 lb of fuel despite a 708 lb increase in the overall weight due to the electrification. Furthermore, Chapman et al. [44] presented an IPTMS design which implements an advanced EPS architecture with AC distribution system and

a conventional TMS. It was found that this IPTMS design can achieve 2.5% reduction in fuel consumption. They also presented an Outer Mould Line (OML) cooling for the same advanced EPS architecture. Compared with the conventional TMS design, the IPTMS design with passive TMS can provide an additional 0.8% reduction in fuel burn. However, it results in an increase in TMS weight. Heersema et al. [42] presented a trade-off study to investigate the impact of the efficiencies of the electrical components on the OML-based passive TMS performance. It was found that, under the low-efficiency condition, the TMS weight is over twice as heavy as the weight at the normal-efficiency condition, and the required area for the installation of the passive TMS exceeds the area available. In [72], Coutinho et al. designed different TMS architectures for a hybrid-electric aircraft that implements either a ram air heat exchanger or a skin heat exchanger for liquid cooling and phase-change cooling. Through the comparison of different TMS architectures, it was found that for more significant improvements in minimising the negative impact of the TMS, the TMS design should be integrated with the EPS, considering different power and design needs. Additionally, the IPTMS design could further reduce fuel consumption when combined with other multidisciplinary analyses, such as those involving aerodynamics and aircraft structures.

The aim of the IPTMS design with advanced EPS and TMS is to minimise the challenges due to the electrical propulsion system, such as the additional weight, the required power for cooling, and the drag due to the use of ram air. The EPS design can affect the requirements of the TMS, while TMS design can similarly contribute to the power requirements. Therefore, trade-off studies are needed between the advancements in thermal and power domains, as the requirements for propulsion and TMSs, operating conditions, and architectures vary for different scales of aircraft. Additionally, the trade-off studies can assess the capability of an IPTMS under different operating conditions. When the power required for cooling constitutes a small proportion of the total power output, it is possible to reduce additional weight by using advanced TMS architectures, or to decrease TMS requirements by employing advanced EPS architectures. If the cooling power constitutes a big proportion of the total power output and affects the battery heat load significantly, it is necessary to manage the power supply strategies of the battery. Therefore, power management optimisation is needed.

4. IPTMS Power Management Optimisation

As presented in Section 2.4, an optimisation strategy for IPTMS can manage thermal and electric energy flows in the aircraft implementing suitable PM strategies. The optimisers aim to (1) reduce operational costs by minimising fuel consumption and battery usage, (2) achieve optimal operations for in-flight components, and (3) satisfy power and thermal demands in different flight conditions. Compared with the conventional optimisers for HEA that optimise power split, the IPTMS optimisers also consider the effect of temperature and auxiliary loads in the TMS. Due to increased electrification, there is a need to manage the power and thermal energies of additional electrical components. Additionally, some TMS components are electrified, such as electrical coolant pumps and cooling fans. These electric components require power supplies and generate heat loads in the aircraft. As a result, it is necessary to assess their performance and optimise their energy flows.

Dynamic programming (DP) is a non-causal power management (PM) strategy. It is highly versatile, enabling it to tackle a wide range of problems [73]. It has been applied for HEA [74–76], with findings indicating that DP can reduce operational costs and improve hydrogen fuel consumption for fuel-cell HEA [75]. Additionally, it also leverages flight scheduling and has nearly no accuracy losses when imitating mechanical operations [74]. However, DP has a disadvantage in computational speed. Pontryagin's Minimum Principle (PMP) is a causal PM strategy derived from DP through variational methods. It can convert the original global optimal problem to a local one [77]. This strategy has been widely studied for HEVs [78]. However, PMP has difficulty solving complex nonlinear problems [79], which presents a challenge for implementing PMP in HEA. Model Predictive Control

(MPC) is an online causal PM strategy. In MPC, the algorithm first predicts the future states based on the system model over a predictive horizon of length N at each sampling time. It then selects the optimal control law from the current time k to the time k + N through rolling optimisation. Once the optimising control law is chosen for the current time k, the results are output only for the time k, disregarding the rest, and the system state is updated. As a result, MPC is more sensitive to the aircraft's future performance. Therefore, while uncertainties in the prediction pose challenges to the robustness of MPC, it is less computationally demanding than DP [80].

Koeln et al. [81] proposed a hierarchical MPC approach for the IPTMS of a military aircraft. They design the IPTMS to provide power for auxiliary loads, emergency loads, and engine starts, while cooling electric components, such as AC/DC loads, advanced electrical equipment, batteries, generators, and inverters. They also assess the dynamic performance of the components with the graph-based approach developed in [82,83]. The MPC is three-level, where the upper level is for planning thermal and power demands, state predictions, and power shedding. The lower-level MPC provides optimal solutions for desired battery power and cooling mass flow rates when satisfying state constraints. The tracking-level MPC tracks the mass flow rates of pumps and fans and makes decisions for valves, inverters, batteries, and power loads. Compared with the baseline controller, the MPC has better performance in capability, safety, and efficiency. A hierarchical MPC approach is implemented for the IPTMS of a UAV in [84]. It was found that the fuel burn can be reduced by 16% with the MPC optimisation strategy compared to the baseline strategy, and the MPC is more reliable. Misley et al. [85] implemented PMP to optimise the performance of the rechargeable battery pack and the TMS for a parallel turboelectric aircraft. The battery is cooled by bleed air extracted from the engine, and it can provide power for propulsion. The optimisation aims to minimise the fuel consumption by controlling the power split (ratio of battery power to total propulsion power) and the cooling ratio (ratio of TMS power required to engine power), while the constraints are the battery temperature and state of charge. The results show that the computational time is reduced from 4 h to 2 min with PMP. However, the fuel consumptions are increased by about 2%.

Both military and civil aircraft need power management optimisation for IPTMS to obtain the optimal battery power supply strategy. When the battery provides required power for TMS and for other power consumers, such as avionics, motors, and anti-icing systems, there could be an optimal solution to manage power split between the battery and the engine-driven generators to minimise the energy consumption. The types of optimisation strategies depend on the requirements, such as computational speed, capability, and stability.

5. IPTMS Modelling

IPTMSs are highly integrated and interdisciplinary, making it beneficial for designers to access the components' performance in various forms, such as in terms of thermal and electrical energies. The models developed for an IPTMS can be used as analysis tools for researchers to investigate the interactions between different subsystems in order to obtain optimal solutions. To evaluate aircraft performance with an IPTMS, the tip-to-tail (T2T) models are developed, usually including the IPTMS, fuel thermal management system, engine system, and air vehicle system. In addition to the aircraft-level models for subsystems, the T2T models also involve component-level models for some critical components. With the integrated models, the T2T models provide a foundation for the aircraft's future optimisation. In some cases, the T2T models are designed for a certain type of aircraft, but the modelling methods are applicable to other aircraft types.

Wolff et al. [86] proposed a modelling and simulation process for IPTMS to maximise the aircraft's energy utilisation while minimising complexity. The component-level models involve electrical components such as the generator, feeders, and actuators. The integrated model has the capability to assess the transient behaviours of the subsystem and find the drivers of the system's performance to define the design points of the subsystems. Through testing, the researchers identified limitations with the assumptions of the design point of the energy storage unit.

Roberts et al. [87–89] developed a generic T2T model, including the subsystem of an air vehicle system, fuel thermal management system, IPTMS, engine system, and control system. They developed the model without proprietary data so that the tool is highly adaptable for different conceptual designs. To increase the model fidelity, they developed transient models for the integrated power package and TMS components (e.g., fuel and oil pumps and heat exchangers). The researchers found that the shortcomings in the TMS design can be detected through the assessment of the transient interactions between the subsystems, which could be neglected in conventional designs. In [90], the T2T model is adapted to a solid oxide fuel cell (SOFC) to analyse the impact of the SOFC at the aircraft level. It was found that the SOFC can improve the overall aircraft efficiency by a small percentage and can reduce the use of bleed air from the main engine. The T2T offers the capability of a trade-off study by evaluating the relationship between subsystems, but only the configuration of the IPTMS integrated with a SOFC is studied.

Due to the multidisciplinary interactions within the subsystems of the IPTMS, a T2T model is required for the IPTMS to assess its performance. A versatile model can be used by different design teams, reducing the operating cost of design. Additionally, the development of IPTMS modelling can contribute to the modular design of IPTMSs.

6. IPTMS Analysis Method

The IPTMS designs in the fields that are mentioned in the previous sections are mostly based on energy analysis relating to the first law of thermodynamics. The energy-based designs aim to minimise aircraft energy usage. However, in 1987, Szargut et al. [91] stated that energy analysis cannot detect most of the problems of thermodynamic imperfection. In 1999, Bejan [92] defined a concept, named "exergy", to measure the quality of various types of energy. The exergy analysis is based on the second law of thermodynamics. A process can satisfy the first law, but compliance does not mean the process is feasible because entropy cannot decrease according to the second law. An exergy analysis is an optimal approach based on the objective functions of entropy generation. Design problems can be identified because aircraft inefficiency can be quantified using exergy analysis. Additionally, researchers can compare the performance of different subsystems with exergy analysis.

Tipton et al. [93] applied an exergy analysis to an IPTMS where the optimisation objectives were exergetic efficiency and total take-off weight. The entropy generated by the IPTMS subsystems is estimated and the total entropy generation of the overall system is found. In this research, the exergy approach can help with the design, but its results are similar to the results of the energy approach and its advantages in IPTMS design need to be validated further. Maser et al. [94] investigated the thermodynamic irreversibility for IPTMSs with exergy approach. They found that exergy optimisation can help to obtain absolute and consistent measurements for aircraft performance trade-off.

In addition to thermodynamic characteristic optimisation, IPTMS optimisation can also achieve light weight at the system level. Rong et al. [95] proposed a system-level analysis method to minimise the total fuel weight penalty for a simplified IPTMS with energy recovery. They adopted the equivalent mass method for analysis and compared the performance of three cooling strategies of the IPTMS. It was found that the optimal cooling strategies are different under different operating conditions.

The development of IPTMS analysis methods can contribute to the trade-off study of an IPTMS. The IPTMS analysis methods can be beneficial for engine cycle designs and lightweight system-level designs. Additionally, the IPTMS analysis methods require validation to ensure their stability.

7. Challenges and Future Opportunities

As presented in previous sections, IPTMSs have been studied widely in different research fields. Owing to increasing electrification, the design of aircraft systems requires a multidisciplinary approach. In this context, TMS designs necessitate integration with other aircraft systems [96], and current IPTMS designs require further enhancement to meet the requirements under various conditions [97]. As a result, it is necessary to discover the current challenges of IPTMSs and their future opportunities.

This section initially presents the challenges and future opportunities for the IPTMS research topics mentioned in Sections 3–6 and explores the potential solutions for civil electrical propulsion aircraft (EPA) based on these discussions.

Regarding the IPTMS architecture design with energy recovery, the system's high integration and multidisciplinary requirements add to its complexity. This complexity brings difficulties in terms of operating cost, reliability, and safety, and presents obstacles for various conceptual designs as well as control and power management system design. This IPTMS design is usually applied to military aircraft due to its ability to reduce overall weight and save volume for other subsystems, thus increasing the aircraft capability. Therefore, future research in this field should focus on integrating more functions into an IPTMS, while enhancing system stability and safety and developing feasible architectures for different aircraft conceptual designs.

As presented in Section 3.2, an IPTMS approach is integrating EPS and TMS architecture designs. It has been found that the advanced EPS architecture design can reduce the requirements of the TMS and ultimately reduce the TMS's overall weight, power, and drag due to ram air. In this research field, the heat loads are usually determined by the efficiencybased method. However, the power required varies across the whole flight mission, which means the current is different as well at different operating points, so the efficiencies of electric components will change. To assess the performance of components and investigate the interactions between the TMS and PMS more accurately, other component modelling methods should be adopted, such as the experiment-based method and the physics-based method. The IPTMSs in this research field are usually designed for HEA. In this case, it is difficult to avoid the use of ram air. Thus, the future research should focus on discovering approaches to minimise ram air usage.

The power management for an IPTMS manages the power and thermal energy flows at the aircraft level. However, its current stage still focuses on the conventional aircraft subsystem architecture. For example, in military aircraft, the power management optimisation is not based on the energy recovery architecture but focuses on the interactions between conventional aircraft subsystems. In civil aircraft, the power management mainly optimises battery performance. Usually, the battery has two power supply strategies: it provides propulsive power only, or provides propulsive power and all the cooling power of the propulsion system. However, there are limited papers that investigate the strategies of battery power supply. For example, the battery can satisfy the required propulsive power and provide from 10% or 20% to 100% of the TMS power needed for propulsion system cooling. Therefore, a potential solution is to design the power management strategies considering the power usage of the TMS to obtain the optimal battery power supply strategies across the aircraft's flight mission.

As presented in Sections 5 and 6, various researchers have developed methods or tools for IPTMSs at the aircraft level, involving modelling, optimisation, and analysis. Although these methods were designed for different power conditions, they need to be more versatile to accommodate other conditions, such as subsonic/supersonic conditions and different aircraft platforms. The integration of more subsystems and the multidisciplinary nature of these methods contribute to their complexity. Therefore, future work should focus on enhancing their versatility and addressing time-related issues (e.g., computational time or simulation time) due to the increasing complexity of design.

Based on the challenges and opportunities presented above, a summary list of requirements for future IPTMSs is as follows:

1. Enhanced system integration: future IPTMSs could benefit from higher levels of integration, not only with TMS and PMS but also with other aircraft subsystems.

- 2. Improved system stability and safety: as the complexity of IPTMSs increase, the importance of system stability and safety may also increase.
- 3. Minimisation of system weight and volume: for military and civil aircraft, future IPTMS research should investigate methods to minimise the system weight and volume.
- 4. Minimisation of cooling power and ram air usage: future research should explore approaches to minimise the use of ram air and the power usage of TMS.
- 5. Optimised power management strategies: Future IPTMS should consider the tradeoff study between propulsive power and the power usage of TMS. This could lead to the development of optimal battery power supply strategies across the aircraft flight mission.
- 6. Investigation of IPTMS optimisation domain: For different operating conditions and different propulsion requirements, the impacts of the TMS and the PMS would be different. Future IPTMSs should investigate at the system level whether the optimisation of the IPTMS should be in the thermal domain or the power domain.
- 7. Versatility and stability of methods and tools: The methods and tools developed for an IPTMS at the aircraft level should be versatile enough to accommodate different power conditions, flight conditions (e.g., subsonic/supersonic, normal/cold-day/hotday conditions), and different aircraft platforms. At the same time, the methods and tools should ensure their stability through validation.
- 8. Addressing time-related issues: with the increasing complexity of design, future work should focus on addressing time-related issues, such as computational time or simulation time.

For civil EPA, the electrified propulsion systems introduce more electrical components than those in military aircraft, thereby increasing the requirements for cooling and energy conservation. Although the IPTMS with energy recovery has the capability to handle cooling and power generation simultaneously, it is not suitable for civil EPA. This is due to this IPTMS typically using fuel as the coolant. While the fuel flow is sufficient for cooling the heat loads of military aircraft, it is inadequate for civil aircraft. Furthermore, the power generation of the IPTMS might not meet the power requirements of civil EPA, as they have more power consumers than military aircraft. In this case, the IPTMS with advanced EPS and TMS could be a feasible IPTMS solution for EPA. With an advanced EPS or TMS design, the thermal and power requirements can be reduced at the architecture level by minimising electric components and simplifying the TMS. Furthermore, an optimal battery power supply strategy can affect the capability of energy saving and the size of the battery, and the battery size then impacts thermal and power requirements. Additionally, the implementation of power management strategy can further optimise the IPTMS performance across the flight mission, reducing energy consumption. Therefore, a potential approach for an IPTMS to address the challenges of EPA is the combination of architecture design (IPTMS with advanced EPS and TMS) and power management optimisation.

Typically, the iteration of civil aircraft models tends to be faster than that of military aircraft. This is because manufacturers need to update their models based on market conditions to meet customer demands and pursue higher profits. Therefore, it is necessary to develop IPTMS platforms with high versatility for civil EPA. This necessitates research on the modelling methods of an IPTMS. The IPTMS platform needs to allow designers to assess the heat and power loads across the flight mission. Additionally, it should also be adaptable when designers decide to use different components. As a result, this platform can enable designers to implement a modular design for IPTMSs. In this context, the future trend in the development of IPTMS analysis methods should be at the system level. This will enable designers to conduct system-level trade-off studies and identify areas for improvement. Given the modular design capability of the platform, these changes can be made easily, thereby enhancing the overall performance of the aircraft in a convenient manner. In this context, the combination of modelling and analysis methods for an IPTMS could be another potential solution for EPA.

8. Conclusions

A comprehensive review on IPTMS in terms of several research areas has been presented to investigate the current challenges and future opportunities of IPTMSs for civil electrical propulsion aircraft (EPA). The current IPTMS research is classified into four research areas: (1) architecture design, (2) power management optimisation, (3) modelling, and (4) analysis method development. Based on the discussions on their challenges and future opportunities, a summary of requirements for future IPTMS design is demonstrated.

The future IPTMS for EPA should deal with the challenges due increasing electrification. Consequently, there are two potential solutions for the IPTMS for EPA:

- The combination of architecture design (IPTMS with advanced EPS and TMS) and power management optimisation can reduce thermal and power requirements and energy consumption.
- 2. The combination of modelling and analysis methods can facilitate the implementation of modular designs and optimise the overall performance of the aircraft.

Author Contributions: Writing—original draft preparation, Z.O.; writing—review and editing and supervision, T.N.; writing—review and editing and supervision, S.J. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflicts of interest.

Abbreviations

- AC Alternating Current
- ACM Air Cycle Machine
- AEA All-Electric Aircraft
- APU Auxiliary Power Unit
- DC Direct Current
- DP Dynamic Programming
- ECS Environmental Control System
- EPA Electrical Propulsion Aircraft
- EPU Emergency Power Unit
- GBx Gearbox
- GT Gas Turbine
- HEA Hybrid-Electric Aircraft
- IPTMS Integrated Power and Thermal Management System
- MEA More Electric Aircraft
- MPC Model Predictive Control
- PAO Polyalphaolephin
- PM Power Management
- PMP Pontryagin's Minimum Principle
- PMS Power Management System
- SOC State of Charge
- T2T Tip-to-Tail
- TMS Thermal Management System

References

- OECD. Air and GHG Emissions. Available online: https://www.oecd-ilibrary.org/environment/air-and-ghg-emissions/ indicator/english_93d10cf7-en (accessed on 24 April 2023).
- Clarke, D.; Flachenecker, F.; Guidetti, E.; Pionnier, P.-A. CO₂ Emissions from Air Transport: A Near-Real-Time Global Database for Policy Analysis. OECD Statistics Working Papers No. 114. 2022. Available online: https://www.oecd-ilibrary.org/economics/ co2-emissions-from-air-transport_ecc9f16b-en (accessed on 15 April 2024).
- United Nations. Sustainable Development Goals. Available online: https://www.un.org/sustainabledevelopment/sustainabledevelopment-goals/ (accessed on 23 August 2022).
- NASA. Strategic Implementation Plan 2017 Update; NASA: Washington, DC, USA, 2017. Available online: https://www.nasa.gov/ sites/default/files/atoms/files/sip-2017-03-23-17-high.pdf (accessed on 24 April 2023).
- 5. European Commission Directorate-General for Research and Innovation. *Fly the Green Deal—Europe's Vision for Sustainable Aviation;* Publications Office of the European Union: Luxemburg, 2022. [CrossRef]
- Schefer, H.; Fauth, L.; Kopp, T.H.; Mallwitz, R.; Friebe, J.; Kurrat, M. Discussion on Electric Power Supply Systems for All Electric Aircraft. *IEEE Access* 2020, *8*, 84188–84216. [CrossRef]
- 7. Sinnett, M. 787 No-bleed Systems: Saving Fuel and Enhancing Operational Efficiencies. Aero Quart. 2007, 18, 6–11.
- Barzkar, A.; Ghassemi, M. Electric Power Systems in More and All Electric Aircraft: A Review. *IEEE Access* 2020, *8*, 169314–169332. [CrossRef]
- Mahefkey, T.; Yerkes, K.; Donovan, B.; Ramalingam, M.L. Thermal Management Challenges for Future Military Aircraft Power Systems. 2004. Available online: https://saemobilus.sae.org/content/2004-01-3204/ (accessed on 24 August 2022).
- Sigthorsson, D.; Oppenheimer, M.W.; Doman, D.B. Aircraft Thermal Endurance Optimization Part I: Using A Mixed Dual Tank Topology and Robust Temperature Regulation. In *AIAA Scitech 2019 Forum*; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2019. [CrossRef]
- Powers, R.W.; Hromisin, S.; McLaughlin, D.K.; Morris, P.J.; Pourhashem, H.; Kalkhoran, I.M.; Lowe, K.T.; Nelson, C.C.; Kan, P.; Lewalle, J.; et al. Dynamic Modelling at a Vehicle Level of a Cryogenic Based Thermal System for a High Powered System. In Proceedings of the 57th AIAA Aerospace Sciences Meeting, San Diego, CA, USA, 7–11 January 2019; Nuzum, S.R., Donovan, A., Roberts, R.A., Wolf, M., Eds.; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2016; pp. 1–15. [CrossRef]
- Aerospace Technology Institute. Accelerating Ambition Foreword Overview of Priorities Air Transport Vision Market Opportunity Vehicles Propulsion and Power Systems Aerostructures Cross-cutting Enablers Economic Impact Closing Remarks. Available online: https://www.ati.org.uk/wp-content/uploads/2021/08/ati-technology-strategy.pdf (accessed on 4 July 2022).
- Ganev, E.; Koerner, M. Power and Thermal Management for Future Aircraft. In Proceedings of the SAE 2013 Aerotech Congress and Exhibition, Montreal, QC, Canada, 24–26 September 2013. [CrossRef]
- 14. Van Heerden, A.S.J.; Judt, D.M.; Jafari, S.; Lawson, C.P.; Nikolaidis, T.; Bosak, D. Aircraft thermal management: Practices, technology, system architectures, future challenges, and opportunities. *Prog. Aerosp. Sci.* **2021**, *128*, 100767. [CrossRef]
- Coutinho, M.; Bento, D.; Souza, A.; Cruz, R.; Afonso, F.; Lau, F.; Suleman, A.; Barbosa, F.R.; Gandolfi, R.; Affonso, W.; et al. A review on the recent developments in thermal management systems for hybrid-electric aircraft. *Appl. Therm. Eng.* 2023, 227, 120427. [CrossRef]
- Asli, M.; König, P.; Sharma, D.; Pontika, E.; Huete, J.; Konda, K.R.; Mathiazhagan, A.; Xie, T.; Höschler, K.; Laskaridis, P. Thermal management challenges in hybrid-electric propulsion aircraft. *Prog. Aerosp. Sci.* 2024, 144, 100967. [CrossRef]
- 17. Zhang, J.; Roumeliotis, I.; Zolotas, A. Sustainable Aviation Electrification: A Comprehensive Review of Electric Propulsion System Architectures, Energy Management, and Control. *Sustainability* **2022**, *14*, 5880. [CrossRef]
- Blaskovic, K. Power Management System for Electric and Hybryd Marine Vessels. In Proceedings of the 2021 44th International Convention on Information, Communication and Electronic Technology (MIPRO), Opatija, Croatia, 27 September–1 October 2021; IEEE: Piscataway, NJ, USA, 2021; pp. 902–907. [CrossRef]
- Isikveren, A.; Kaiser, S.; Pornet, C.; Vratny, P. Pre-design strategies and sizing techniques for dual-energy aircraft. *Aircr. Eng. Aerosp. Technol.* 2014, *86*, 525–542. [CrossRef]
- Brelje, B.J.; Martins, J.R.R.A. Electric, hybrid, and turboelectric fixed-wing aircraft: A review of concepts, models, and design approaches. Prog. Aerosp. Sci. 2019, 104, 1–19. [CrossRef]
- Alrashed, M.; Nikolaidis, T.; Pilidis, P.; Jafari, S. Utilisation of turboelectric distribution propulsion in commercial aviation: A review on NASA's TeDP concept. *Chin. J. Aeronaut.* 2021, 34, 48–65. [CrossRef]
- Xie, Y.; Savvarisal, A.; Tsourdos, A.; Zhang, D.; Gu, J. Review of hybrid electric powered aircraft, its conceptual design and energy management methodologies. *Chin. J. Aeronaut.* 2020, 34, 432–450. [CrossRef]
- 23. Sahoo, S.; Zhao, X.; Kyprianidis, K. A Review of Concepts, Benefits, and Challenges for Future Electrical Propulsion-Based Aircraft. *Aerospace* 2020, 7, 44. [CrossRef]
- 24. Larminie, J.; Lowry, J. Electric Machines and their Controllers. In *Electric Vehicle Technology Explained*; Wiley: Hoboken, NJ, USA, 2012; p. 145. [CrossRef]
- Gkoutzamanis, V.G.; Tsentis, S.E.; Mylonas, O.S.; Kalfas, A.I.; Kyprianidis, K.G.; Tsirikoglou, P.; Sielemann, M. Thermal Management System Considerations for a Hybrid-Electric Commuter Aircraft. J. Thermophys. Heat Transf. 2022, 36, 650–666. [CrossRef]

- Pal, D.; Severson, M. Liquid Cooled System for Aircraft Power Electronics Cooling. In Proceedings of the 16th IEEE Intersociety Conference on Thermal and Thermomechanical Phenomena in Electronic Systems (ITherm), Orlando, FL, USA, 30 May–2 June 2017; IEEE: Piscataway, NJ, USA, 2017; pp. 800–805. [CrossRef]
- 27. Jafari, S.; Nikolaidis, T. Thermal Management Systems for Civil Aircraft Engines: Review, Challenges and Exploring the Future. *Appl. Sci.* 2018, *8*, 2044. [CrossRef]
- Roumeliotis, I.; Castro, L.; Jafari, S.; Pachidis, V.; De Riberolles, L.; Broca, O.; Unlu, D. Integrated Systems Simulation for Assessing Fuel Thermal Management Capabilities for Hybrid-Electric Rotorcraft. In Proceedings of the ASME Turbo Expo 2020: Turbomachinery Technical Conference and Exposition, Virtual, 21–25 September 2020. [CrossRef]
- Jafari, S.; Bouchareb, A.; Nikolaidis, T. Thermal Performance Evaluation in Gas Turbine Aero Engines Accessory Gearbox. Int. J. Turbomach. Propuls. Power 2020, 5, 21. [CrossRef]
- Liu, H.; Wei, Z.; He, W.; Zhao, J. Thermal issues about Li-ion batteries and recent progress in battery thermal management systems: A review. *Energy Convers. Manag.* 2017, 150, 304–330. [CrossRef]
- Affonso, W.; Gandolfi, R.; dos Reis, R.J.N.; da Silva, C.R.I.; Rodio, N.; Kipouros, T.; Laskaridis, P.; Chekin, A.; Ravikovich, Y.; Ivanov, N.; et al. Thermal Management challenges for HEA-FUTPRINT 50. *IOP Conf. Ser. Mater. Sci. Eng.* 2021, 1024, 012075. [CrossRef]
- U.S. Department of Energy. EERE Success Story—Supercharging Concentrating Solar Power Plant Performance. 2017. Available online: https://www.energy.gov/eere/success-stories/articles/eere-success-story-supercharging-concentrating-solar-powerplant (accessed on 26 April 2023).
- Anselmi, E.; Bunce, I.; Pachidis, V. An Overview of Initial Operational Experience with the Closed-Loop sCO₂ Test Facility at Cranfield University. In Proceedings of the ASME Turbo Expo 2019: Turbomachinery Technical Conference and Exposition, Phoenix, AZ, USA, 17–21 June 2019. [CrossRef]
- 34. Eastman Chemical Company. Heat Transfer Fluids. 2020. Available online: https://www.therminol.com/heat-transfer-fluids (accessed on 26 April 2023).
- 35. Ravikumar, R.; Herbert, G.M.J.; Binoy, L.; Ratheesh, R.; Krishnakumar, T.S.; Indulal, C.R. Experimental investigations on the thermal properties of rare earth-metal oxide based therminol nano fluids for heat transfer applications. *Sādhanā* **2023**, *48*, 62. [CrossRef]
- 36. Merzvinskas, M.; Bringhenti, C.; Tomita, J.; de Andrade, C. Air conditioning systems for aeronautical applications: A review. *Aeronaut. J.* **2019**, *124*, 499–532. [CrossRef]
- 37. Pérez-Grande, I.; Leo, T.J. Optimization of a commercial aircraft environmental control system. *Appl. Therm. Eng.* 2002, 22, 1885–1904. [CrossRef]
- Ahlers, M.F. Aircraft Thermal Management. In *Encyclopedia of Aerospace Engineering*; John Wiley & Sons: Chichester, UK, 2011. [CrossRef]
- Rheaume, J.M.; MacDonald, M.; Lents, C.E. Commercial Hybrid Electric Aircraft Thermal Management System Design, Simulation, and Operation Improvements. In Proceedings of the AIAA Propulsion and Energy 2019 Forum, Indianapolis, IN, USA, 19–22 August 2019. [CrossRef]
- Annapragada, R.; Sur, A.; Mahmoudi, R.; Macdonald, M.; Lents, C.E. Hybrid Electric Aircraft Battery Heat Acquisition System. In Proceedings of the 2018 AIAA/IEEE Electric Aircraft Technologies Symposium, Cincinnati, OH, USA, 12–14 July 2018; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2018. [CrossRef]
- 41. Kellermann, H.; Habermann, A.L.; Vratny, P.C.; Hornung, M. Assessment of fuel as alternative heat sink for future aircraft. *Appl. Therm. Eng.* **2020**, *170*, 114985. [CrossRef]
- Heersema, N.; Jansen, R.; Kiris, C.C.; Chau, T.; Machado, L.M.; Duensing, J.C.; Mirhashemi, A.; Chapman, J.; French, B.D.; Miller, L.; et al. Thermal Management System Trade Study for SUSAN Electrofan Aircraft. In *AIAA SCITECH 2022 Forum*; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2022. [CrossRef]
- Sozer, E.; Maldonado, D.; Bhamidapati, K.; Schnulo, S.L. Computational Evaluation of an OML-based Heat Exchanger Concept for HEATheR. In *AIAA Propulsion and Energy 2020 Forum*; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2020. [CrossRef]
- 44. Schnulo, S.L.; Chapman, J.W.; Hanlon, P.; Hasseeb, H.; Jansen, R.; Sadey, D.; Sozer, E.; Jensen, J.; Maldonado, D.; Bhamidapati, K.; et al. Assessment of the Impact of an Advanced Power System on a Turboelectric Single-Aisle Concept Aircraft. In *AIAA Propulsion and Energy 2020 Forum*; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2020; pp. 1–18. [CrossRef]
- 45. Wang, T.; Tseng, K.; Zhao, J.; Wei, Z. Thermal investigation of lithium-ion battery module with different cell arrangement structures and forced air-cooling strategies. *Appl. Energy* **2014**, *134*, 229–238. [CrossRef]
- Simon, D.L.; Connolly, J.W.; Culley, D.E. Control Technology Needs for Electrified Aircraft Propulsion Systems. J. Eng. Gas Turbines Power 2019, 142, 11025. [CrossRef]
- Olatomiwa, L.; Mekhilef, S.; Ismail, M.; Moghavvemi, M. Energy management strategies in hybrid renewable energy systems: A review. *Renew. Sustain. Energy Rev.* 2016, 62, 821–835. [CrossRef]
- 48. Gong, X.; Wang, J.; Ma, B.; Lu, L.; Hu, Y.; Chen, H. Real-Time Integrated Power and Thermal Management of Connected HEVs Based on Hierarchical Model Predictive Control. *IEEE/ASME Trans. Mechatron.* **2021**, *26*, 1271–1282. [CrossRef]
- Cinar, G.; Mavris, D.N.; Emeneth, M.; Schneegans, A.; Fefermann, Y. Development of Parametric Power Generation and Distribution Subsystem Models at the Conceptual Aircraft Design Stage. In Proceedings of the 55th AIAA Aerospace Sciences Meeting, Grapevine, TX, USA, 9–13 January 2017. [CrossRef]

- 50. Brahma, A.; Guezennec, Y.; Rizzoni, G. Optimal energy management in series hybrid electric vehicles. In Proceedings of the 2000 American Control Conference, Chicago, IL, USA, 28–30 June 2000; Volume 1, pp. 60–64. [CrossRef]
- Wang, R.; Lukic, S.M. Dynamic Programming Technique in Hybrid Electric Vehicle Optimization. In Proceedings of the 2012 IEEE International Electric Vehicle Conference, Greenville, SC, USA, 4–8 March 2012; IEEE: Piscataway, NJ, USA, 2012; pp. 1–8. [CrossRef]
- 52. Chen, Z.; Mi, C.C.; Xu, J.; Gong, X.; You, C. Energy Management for a Power-Split Plug-in Hybrid Electric Vehicle Based on Dynamic Programming and Neural Networks. *IEEE Trans. Veh. Technol.* **2014**, *63*, 1567–1580. [CrossRef]
- 53. Pérez, L.V.; Bossio, G.R.; Moitre, D.; García, G.O. Optimization of power management in an hybrid electric vehicle using dynamic programming. *Math. Comput. Simul.* **2006**, *73*, 244–254. [CrossRef]
- 54. Gao, J.-P.; Zhu, G.-M.G.; Strangas, E.G.; Sun, F.-C. Equivalent fuel consumption optimal control of a series hybrid electric vehicle. *Proc. Inst. Mech. Eng. Part D J. Automob. Eng.* **2009**, 223, 1003–1018. [CrossRef]
- Sciarretta, A.; Back, M.; Guzzella, L. Optimal Control of Parallel Hybrid Electric Vehicles. *IEEE Trans. Control Syst. Technol.* 2004, 12, 352–363. [CrossRef]
- Pisu, P.; Rizzoni, G. A Comparative Study Of Supervisory Control Strategies for Hybrid Electric Vehicles. *IEEE Trans. Control Syst. Technol.* 2007, 15, 506–518. [CrossRef]
- Musardo, C.; Rizzoni, G.; Guezennec, Y.; Staccia, B. A-ECMS: An Adaptive Algorithm for Hybrid Electric Vehicle Energy Management. *Eur. J. Control* 2005, 11, 509–524. [CrossRef]
- Tang, L.; Rizzoni, G.; Onori, S. Energy Management Strategy for HEVs Including Battery Life Optimization. *IEEE Trans. Transp. Electrif.* 2015, 1, 211–222. [CrossRef]
- 59. Hongli, Z.; Yu, H.; Liang, C. Experimental study on a small Brayton air refrigerator under –120 °C. *Appl. Therm. Eng.* **2009**, *29*, 1702–1706. [CrossRef]
- 60. Kabir, R.; Kaddoura, K.; McCluskey, P.; Kizito, J.P. Investigation of a Cooling System for a Hybrid Airplane. In 2018 AIAA/IEEE Electric Aircraft Technologies Symposium; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2018. [CrossRef]
- 61. Yu, S.; Ganev, E. Next Generation Power and Thermal Management System. *SAE Int. J. Aerosp.* **2008**, *1*, 1107–1121. [CrossRef]
- 62. Lui, C.; Dooley, M. *Electric Thermal Management Architectures*; SAE Technical Paper Series; SAE International: Warrendale, PA, USA, 2013. [CrossRef]
- 63. Newman, R.W.; Dooley, M.; Lui, C. Efficient propulsion, power, and thermal management integration. In Proceedings of the 49th AIAA/ASME/SAE/ASEE Joint Propulsion Conference, Virtual, 15 July 2013; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2013. [CrossRef]
- 64. Abolmoali, P.; Parrilla, J.A.; Hamed, A. Integrated Aircraft Thermal Management & Power Generation: Reconfiguration of a Closed Loop Air Cycle System as a Brayton Cycle Gas Generator to Support Auxiliary Electric Power Generation; SAE Technical Papers; SAE International: Warrendale, PA, USA, 2014. [CrossRef]
- Payne, N.; Wolff, M.; Roberts, R.; Elston, L.; McCoppin, J. Experimental Validation of a Combined Thermal Management and Power Generation System using a MultiMode Rankine Cycle. In *AIAA Propulsion and Energy 2020 Forum*; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2020; pp. 1–17. [CrossRef]
- 66. Stoia, M.F.; Ek, G.W.; Bowcutt, K.G.; Needels, J.T. Integrated Power and Thermal Management System for High Speed Aircraft. In *AIAA Propulsion and Energy 2021 Forum*; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2021. [CrossRef]
- 67. Rheaume, J.; Lents, C.E. Design and Simulation of a Commercial Hybrid Electric Aircraft Thermal Management System. In Proceedings of the 2018 AIAA/IEEE Electric Aircraft Technologies Symposium, Cincinnati, OH, USA, 12–14 July 2018; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2018. [CrossRef]
- Kim, J.; Kwon, K.; Roy, S.; Garcia, E.; Mavris, D.N. Megawatt-Class Turboelectric Distributed Propulsion, Power, and Thermal SYSTEMS for aircraft. In Proceedings of the 2018 AIAA Aerospace Sciences Meeting, Kissimmee, FL, USA, 8–12 January 2018; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2018. [CrossRef]
- Chapman, J.W.; Hasseeb, H.; Schnulo, S.L. Thermal Management System Design for Electrified Aircraft Propulsion Concepts. In AIAA Propulsion and Energy 2020 Forum; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2020; pp. 1–23. [CrossRef]
- Thomas, G.L.; Chapman, J.W.; Hasseeb, H.; Alencar, J.F.; Sadey, D.; Csank, J. Multidisciplinary Systems Analysis of a Six Passenger Quadrotor Urban Air Mobility Vehicle Powertrain. In *AIAA Propulsion and Energy 2020 Forum*; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2020; pp. 1–22. [CrossRef]
- Abolmoali, P.C.; Donovan, A.B.; Patnaik, S.S.; McCarthy, P.; Dierker, D.; Jones, N.; Buettner, R. Integrated Propulsive and Thermal Management System Design for Optimal Hybrid Electric Aircraft Performance. In *AIAA Propulsion and Energy 2020 Forum*; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2020; pp. 1–17. [CrossRef]
- 72. Coutinho, M.; Afonso, F.; Souza, A.; Bento, D.; Gandolfi, R.; Barbosa, F.R.; Lau, F.; Suleman, A. A Study on Thermal Management Systems for Hybrid–Electric Aircraft. *Aerospace* 2023, 10, 745. [CrossRef]
- Larsson, V.; Johannesson, L.; Egardt, B. Analytic Solutions to the Dynamic Programming Subproblem in Hybrid Vehicle Energy Management. *IEEE Trans. Veh. Technol.* 2015, 64, 1458–1467. [CrossRef]
- 74. Sergent, A.; Ramunno, M.; D'Arpino, M.; Canova, M.; Perullo, C. Optimal Sizing and Control of Battery Energy Storage Systems for Hybrid Turboelectric Aircraft. *SAE Int. J. Adv. Curr. Pract. Mobil.* **2020**, *2*, 1266–1278. [CrossRef]

- 75. Fares, D.; Chedid, R.; Panik, F.; Karaki, S.; Jabr, R. Dynamic programming technique for optimizing fuel cell hybrid vehicles. *Int. J. Hydrogen Energy* **2015**, *40*, 7777–7790. [CrossRef]
- Leite, J.P.S.P.; Voskuijl, M. Optimal energy management for hybrid-electric aircraft. *Aircr. Eng. Aerosp. Technol.* 2020, 92, 851–861. [CrossRef]
- Hou, C.; Ouyang, M.; Xu, L.; Wang, H. Approximate Pontryagin's minimum principle applied to the energy management of plugin hybrid electric vehicles. *Appl. Energy* 2014, 115, 174–189. [CrossRef]
- 78. Onori, S.; Serrao, L.; Rizzoni, G. Hybrid Electric Vehicles; Springer Science and Business Media LLC: Dordrecht, The Netherlands, 2016.
- 79. Nguyen, B.-H.; Vo-Duy, T.; Ta, M.C.; Trovao, J.P.F. Optimal Energy Management of Hybrid Storage Systems Using an Alternative Approach of Pontryagin's Minimum Principle. *IEEE Trans. Transp. Electrif.* **2021**, *7*, 2224–2237. [CrossRef]
- Hu, Q.; Amini, M.R.; Wang, H.; Kolmanovsky, I.; Sun, J. Integrated Power and Thermal Management of Connected HEVs via Multi-Horizon MPC. In Proceedings of the 2020 American Control Conference (ACC), Denver, CO, USA, 1–3 July 2020; IEEE: Piscataway, NJ, USA, 2020; pp. 3053–3058. [CrossRef]
- Koeln, J.P.; Pangborn, H.C.; Williams, M.A.; Kawamura, M.L.; Alleyne, A.G. Hierarchical Control of Aircraft Electro-Thermal Systems. *IEEE Trans. Control Syst. Technol.* 2019, 28, 1218–1232. [CrossRef]
- Pangborn, H.C.; Williams, M.A.; Koeln, J.P.; Alleyne, A.G. Graph-Based Hierarchical Control of Thermal-Fluid Power Flow Systems. In Proceedings of the American Control Conference, Seattle, WA, USA, 24–26 May 2017; IEEE: Piscataway, NJ, USA, 2017; pp. 2099–2105. [CrossRef]
- 83. Williams, M.A.; Koeln, J.P.; Pangborn, H.C.; Alleyne, A.G. Dynamical Graph Models of Aircraft Electrical, Thermal, and Turbomachinery Components. J. Dyn. Syst. Meas. Control 2018, 140, 041013. [CrossRef]
- Aksland, C.T.; Tannous, P.J.; Wagenmaker, M.J.; Pangborn, H.C.; Alleyne, A.G. Hierarchical Predictive Control of an Unmanned Aerial Vehicle Integrated Power, Propulsion, and Thermal Management System. *IEEE Trans. Control Syst. Technol.* 2023, 31, 1280–1295. [CrossRef]
- Misley, A.A.; D'Arpino, M.; Ramesh, P.; Canova, M. A real-time energy management strategy for hybrid electric aircraft propulsion systems. In *AIAA Propulsion and Energy 2021 Forum*; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2021. [CrossRef]
- 86. Wolff, M. Integrated Thermal/Power/Propulsion/Vehicle Modeling Issues Related to a More Electric Aircraft Architecture. Wright-Patterson AFB: NATO. 2010; p. 1–26. Available online: https://www.sto.nato.int/publications/STO%20Meeting%20 Proceedings/RTO-MP-AVT-178/MP-AVT-178-28.pdf (accessed on 9 December 2021).
- Roberts, R.; Eastbourn, S.; Maser, A. Generic aircraft thermal tip-to-tail modeling and simulation. In Proceedings of the 47th AIAA/ASME/SAE/ASEE Joint Propulsion Conference and Exhibit, San Diego, CA, USA, 31 July–3 August 2011; American Institute of Aeronautics and Astronautics: Reston, VA, USA, 2011. [CrossRef]
- 88. Roberts, R.; Eastbourn, S. Vehicle Level Tip-to-Tail Modeling of an Aircraft. Int. J. Thermodyn. 2014, 17, 107–115. [CrossRef]
- 89. Roberts, R.A.; Decker, D.D. Control Architecture Study Focused on Energy Savings of an Aircraft Thermal Management System. *J. Dyn. Syst. Meas. Control* **2014**, *136*, 041003. [CrossRef]
- Roberts, R.A.; Wolff, M.; Nuzum, S.; Donovan, A. Assessment of the Vehicle Level Impact for a SOFC Integrated with the Power and Thermal Management System of an Air Vehicle. In Proceedings of the ASME 2015 Dynamic Systems and Control Conference, Columbus, OH, USA, 28–30 October 2015; American Society of Mechanical Engineers: New York, NY, USA, 2015. [CrossRef]
- 91. Szargut, J.; Morris, D.R.; Steward, F.R. Exergy Analysis of Thermal, Chemical, and Metallurgical Processes; Hemisphere Publishing Corporation: Washington, DC, USA, 1987.
- 92. Bejan, A. A Role for Exergy Analysis and Optimization in Aircraft Energy-System Design. In *Advanced Energy Systems;* American Society of Mechanical Engineers: New York, NY, USA, 1999; pp. 209–217. [CrossRef]
- 93. Figliola, R.S.; Tipton, R.; Li, H. Exergy Approach to Decision-Based Design of Integrated Aircraft Thermal Systems. J. Aircr. 2003, 40, 49–55. [CrossRef]
- 94. Maser, A.; Garcia, E.; Mavris, D. Characterization of thermodynamic irreversibility for integrated propulsion and thermal management systems design. In Proceedings of the 50th AIAA Aerospace Sciences Meeting Including the New Horizons Forum and Aerospace Exposition, Nashville, TN, USA, 9–11 January 2012. [CrossRef]
- 95. Rong, A.; Pang, L.; Jiang, X.; Qi, B.; Shi, Y. Analysis and comparison of potential power and thermal management systems for high-speed aircraft with an optimization method. *Energy Built Environ.* **2021**, *2*, 13–20. [CrossRef]
- 96. Walters, E.; Amrhein, M.; O'Connell, T.; Iden, S.; Lamm, P.; Yerkes, K.; Wolff, M.; McCarthy, K.; Raczkowski, B.; Wells, J.; et al. INVENT Modeling, Simulation, Analysis and Optimization. In Proceedings of the 48th AIAA Aerospace Sciences Meeting Including the New Horizons Forum and Aerospace Exposition, Orlando, FL, USA, 4–7 January 2010. [CrossRef]
- 97. Dooley, M.; Lui, N.; Newman, R.; Lui, C. Aircraft Thermal Management—Heat Sink Challenge. In Proceedings of the SAE 2014 Aerospace Systems and Technology Conference, Cincinnati, OH, USA, 23–25 September 2014. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.